
Daily Status 10-23-2016
Clear, cool, breezy.

UDel network down yesterday. Things are mostly up now, but eddy cannot do http/ftp requests to the outside.  UPD data coming in though.

To do:

Reshoot boom angles at site 8 and 15 where we used tripod for stabilizing. We think tripod could be affecting the compass. (must be done at low 
tide, so this has to wait.
Shoot sonics on tower.

Ongoing:

Watching main tower wdir data and forecasts to determine appropriate conditions to have wind turbine turned off (at the request of the PI). She's 
looking for ENE winds between 40-80deg greater than 5m/s

T/RH: 42m  TRH reporting warmer and drier than at other levels. 
P: ok
2D Gill: ok

csat u,v:  ok
csat ldiag: ok
csat w, tc: ok
3D RMY: ok
EC150: ok
motes: ok
Wetness: ok
radiation: ok
Tsoil: ok
Gsoil: ok
Qsoil: Qsoil.5cm.m1.s9 reporting intermittently
Cvsoil: ok
Rainr: ok.
Vbatt: ok
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